% These are my slides for my ASA talk about my research.

%\documentclass{slides}

\documentclass{seminar}
%\documentclass[article]{seminar}
\special{landscape}
%\twoup[1]
\pagestyle{empty}
%\onlyslides{28,29}

\input epsf.tex

\usepackage{amsbsy}

\newcommand{\bp}{\mbox{\boldmath $\phi$}}
\newcommand{\bT}{\mbox{\boldmath $\Theta$}}

\newcommand{\by}{\ensuremath {\bf y}}
\newcommand{\ba}{\ensuremath {\bf a}}
\newcommand{\bs}{\ensuremath {\bf s}}
\newcommand{\be}{\ensuremath {\bf e}}
\newcommand{\bY}{\ensuremath {\bf Y}}
\newcommand{\bS}{\ensuremath {\bf S}}
\newcommand{\bE}{\ensuremath {\bf E}}

\newcommand{\bt}{\ensuremath {\bf \tau}}
\newcommand{\btheta}{\ensuremath {\bf \theta}}

\newcommand{\bvep}{\mbox{\boldmath $\varepsilon$}}

\newcommand{\noisee}{\ensuremath {\sigma^2}}
\newcommand{\noiseep}{\ensuremath {c\sigma^2}}
\newcommand{\pree}{\ensuremath {\tau}}
\newcommand{\preep}{\ensuremath {\tau}}

\newcommand{\tc}{\ensuremath {\left(\frac{\tau}{c}\right)}}

\newcommand{\sumone}{\ensuremath {\sum_{t=max(i,l+1)}^{min(i+m,n)}}}
\newcommand{\sumtwo}{\ensuremath {\sum_{t=i}^{min(i+p,n)}}}

\begin{document}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% Ch. 4

\begin{slide}
\begin{figure}[!htb]
  \epsfxsize=5.5in
  %\epsfysize=8in
  \centerline{\epsfbox{Pny.ps}}
  \caption{$P_n$-phase Event 054: Scaled data, ${\by}_k$, $k=1,...,q$, where $q=5$ and $n=220$.}
  \label{fig:P054y}
\end{figure}
\end{slide}

\begin{slide}
\begin{figure}[!htb]
  \epsfxsize=5.5in
  %\epsfysize=8in
  \centerline{\epsfbox{PnaEst.eps}}
  \caption{$P_n$-phase Event 054: Posterior estimates of the {\it pulse sequence},
  $\hat{a}_j$, $j=1,...,m$, where $m = 40$.}
  \label{fig:P054aEst}
\end{figure}
\end{slide}

\begin{slide}
\begin{figure}[!htb]
  \epsfxsize=5.5in
  %\epsfysize=8in
  \centerline{\epsfbox{PnsEst.ps}}
  \caption{$P_n$-phase Event 054: Posterior estimates of the {\it underlying signal-path effects},
  $\hat{\bs}_k$, on each channel.}
  \label{fig:P054sEst}
\end{figure}
\end{slide}

\begin{slide}
\begin{figure}[!htb]
  \epsfxsize=5.5in
  %\epsfysize=8in
  \centerline{\epsfbox{Pnyyhat2.ps}}
  \caption{$P_n$-phase Event 054: Solid: Scaled data, ${\by}_k$.
  Dots: Posterior estimates, $\hat{\by}_k$, shifted one point to the left.}
  \label{fig:PnyyEst2}
\end{figure}
\end{slide}

\begin{slide}
\begin{figure}[!htb]
  \epsfxsize=5.5in
  %\epsfysize=8in
  \centerline{\epsfbox{PMCa1.ps}}
  \caption{$P_n$-phase Event 054: Last 5,000 iterations in the
  Markov chains produced for the peaks in $\hat{a}_j$, $j=a,...,m$.}
  \label{fig:PnMCa1}
\end{figure}
\end{slide}

\begin{slide}
\begin{figure}[!htb]
 \epsfxsize=5.5in
  %\epsfysize=8in
  \centerline{\epsfbox{PMCa2.ps}}
  \caption{$P_n$-phase Event 054: Last 5,000 iterations in the
  Markov chains produced for the peaks in $\hat{a}_j$, $j=a,...,m$.}
  \label{fig:PnMCa2}
\end{figure}
\end{slide}

\begin{slide}
\begin{figure}[!htb]
  \epsfxsize=5.5in
  %\epsfysize=8in
  \centerline{\epsfbox{PPosta.ps}}
  \caption{$P_n$-phase Event 054: Posteriors of $a_j$, $j=1,...,m$.}
  \label{fig:PnPosta}
\end{figure}
\end{slide}

\begin{slide}
\begin{figure}[!htb]
  \epsfxsize=5.5in
  %\epsfysize=8in
  \centerline{\epsfbox{PMCphi.eps}}
  \caption{$P_n$-phase Event 054: Markov chains for $\bp$.}
  \label{fig:PnMCphi}
\end{figure}
\end{slide}

\begin{slide}
\begin{figure}[!htb]
  \epsfxsize=5.5in
  %\epsfysize=8in
  \centerline{\epsfbox{PMCet1.eps}}
  \caption{$P_n$-phase Event 054: Markov chains for $\eta$ and $\tau$.}
  \label{fig:PnMCet}
\end{figure}
\end{slide}

\begin{slide}
\begin{figure}[!htb]
  \epsfxsize=5.5in
  %\epsfysize=8in
  \centerline{\epsfbox{PPostpet.ps}}
  \caption{$P_n$-phase Event 054: Posteriors of $\phi_1$, $\phi_2$, $\phi_3$, $\eta$, and $\tau$.}
  \label{fig:PnPostp}
\end{figure}
\end{slide}

\begin{slide}
\begin{figure}[!htb]
  \epsfxsize=5.5in
  %\epsfysize=8in
  \centerline{\epsfbox{Pgy.ps}}
  \caption{$P_g$-phase Event 054: Scaled data, $y_k(t)$, $k=1,...,q$, $t=1,...,n$, where $q=5$ and $n=220$.}
  \label{fig:Pg054y}
\end{figure}
\end{slide}

\begin{slide}
\begin{figure}[!htb]
  \epsfxsize=5.5in
  %\epsfysize=8in
  \centerline{\epsfbox{PgaEst1.eps}}
  \caption{$P_g$-phase Event 054: Posterior estimates of the {\it pulse sequence},
  $\hat{a}_j$, $j=1,...,m$, where $m = 40$.}
  \label{fig:Pg054aEst}
\end{figure}
\end{slide}

\begin{slide}
\begin{figure}[!htb]
  \epsfxsize=5.5in
  %\epsfysize=8in
  \centerline{\epsfbox{Sny.ps}}
  \caption{$S_n$-phase Event 054: Scaled data, $y_k(t)$, $k=1,...,q$, $t=1,...,n$, where $q=5$ and $n=256$.}
  \label{fig:Sn054y}
\end{figure}
\end{slide}

\begin{slide}
\begin{figure}[!htb]
  \epsfxsize=5.5in
  %\epsfysize=8in
  \centerline{\epsfbox{SnaEst1.eps}}
  \caption{$S_n$-phase Event 054: Posterior estimates of the {\it pulse sequence},
  $\hat{a}_j$, $j=1,...,m$, where $m = 40$.}
  \label{fig:Sn054aEst}
\end{figure}
\end{slide}

\begin{slide}
\begin{figure}[!htb]
 \epsfxsize=5.5in
  %\epsfysize=8in
  \centerline{\epsfbox{Lgy.ps}}
  \caption{$L_g$-phase Event 054: Scaled data, $y_k(t)$, $k=1,...,q$, $t=1,...,n$, where $q=5$ and $n=512$.}
  \label{fig:L054y}
\end{figure}
\end{slide}

\begin{slide}
\begin{figure}[!htb]
  \epsfxsize=5.5in
  %\epsfysize=8in
  \centerline{\epsfbox{Lga.eps}}
  \caption{$L_g$-phase Event 054: Posterior estimates of the {\it pulse sequence},
  $\hat{a}_j$, $j=1,...,m$, where $m = 50$.}
  \label{fig:L054aEst}
\end{figure}
\end{slide}

%%% Comparison Slide %%%

\begin{slide}
\begin{figure}[!htb]
  \epsfxsize=5.5in
  %\epsfysize=8in
  \centerline{\epsfbox{aEst054.ps}}
  \caption{Event 054: Posterior estimates of the {\it pulse sequence} from each phase.}
  \label{fig:ALLaEst}
\end{figure}
\end{slide}


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\end{document}
